Author’s response to reviews

Title: How to conserve dignity in palliative care: Suggestions from older patients, significant others, and health care professionals in Swedish municipal care

Authors:

Ulrika Östlund (ulrika.ostlund@regiongavleborg.se)
Karin Blomberg (karin.blomberg@oru.se)
Annika Söderman (annika.soderman@oru.se)
Carina Werkander Harstäde (carina.harstade@lnu.se)

Version: 3 Date: 09 Jan 2019

Author’s response to reviews:

Dear Editor thank you for your quick and positive reply, we are delighted.

The following amendments have been made to the manuscript.

- Page 23 line 533: Abbreviations – the abbreviation HPs: Health care professionals has been added
- Page 26 line 586: Reference number 7 – Lindstrom UA has been corrected to Lindström UÅ
- Page 26 line 588: Reference number 8 – the journal name has been abbreviated (J Med Ethics) and changed from 2010;36(3):160-4 to 2010;36:160-4.
- Page 26 line 589: Reference number 9 – Rynnerstrom has been corrected to Rytterström
- Page 26 line 597: Reference number 12 – page number changed from 635-643 to 635-43
- Page 27 line 617: Reference number 21 - page number changed from 666-672 to 666-72
- Page 27 line 615: Reference number 20 - the journal name has been abbreviated (Soc Sci Med)
- Page 28 line 640: Reference number 30 – Kyngas has been corrected to Kyngås
- Page 28 line 656: Reference number 38 - corrected from Lützen K, to Lützén K.

Sincerely Ulrika Östlund, corresponding author